Author's response to reviews

Title: Nursing Uniform wearing practice among Nurses working in Northwest Ethiopia: a cross-sectional institution based Study.

Authors:

Etaferahu Alamaw Desta EAD (tafealamaw@gmail.com)
Berihun Assefa Dachew BAD (berihunassefa21@gmail.com)
Mignote Hailu Gebrie MHG (elatman.hail@gmail.com)

Version: 2 Date: 26 March 2015

Author's response to reviews:

Dear Editors;
Greetings;
Thank you very much for your comment;
We inserted the page number and line numbering in the revised version of the manuscript. However, since we do not use any type of abbreviation in the main document; there is no list of abbreviation section.

Thank you;